# **Hall D Controls Meeting 24-Jan-2019**

Date: 01/24/19 Time: 10:00am

<u>Attendees</u>: Hovanes Egiyan, Brian Eng, Amanda Hoebel, Tyler Lemon, Nick Sandoval, Scot Spiegel, Tim Whitlatch, Beni Zihlmann

#### 1. General:

- 1.1. Hall will close on Tuesday.
  - 1.1.1. Will possibly re-open the following Thursday.

### 2. Solenoid status:

2.1. Checkout procedure is being conducted.

#### 3. ComCal:

- 3.1. Around 7am, temperature read -5°C in ComCal.
  - 3.1.1. Will be investigated.

## 4. DIRC controls:

- 4.1. North optical boards delayed until summer.
- 4.2. Two MPOD controllers with possible firmware issue.
  - 4.2.1. Controller could not turn on crate interlocks.
  - 4.2.2. Updating firmware caused communication issues with backplane.
  - 4.2.3. Firmware will be changed until the right one is found to fix problems.
- 4.3. Voltage channels disabled.
  - 4.3.1. One slot not reading back on card.
- 4.4. Optical box flow and liquid level not in Environment GUI in EPICS.
- 4.5. Alarms to be added to GUIs.